Author's response to reviews

Title: Sudden Infant Death Syndrome and Prenatal Maternal Smoking: Rising Attributed Risk in the Back to Sleep Era

Authors:

Mark E Anderson (manderso@dhha.org)
Daniel C Johnson (daniel.johnson@uchsc.edu)
Holly A Batal (hbatal@dhha.org)

Version: 4 Date: 12 November 2004

Author's response to reviews: see over
“Sudden Infant Death Syndrome and Prenatal Maternal Smoking: Rising Attributed Risk in the Back to Sleep Era”

November 12, 2004

Dear Editorial Staff:

Thank you for your continuing efforts leading toward publication of this manuscript. I’m pleased you find it suitable for BMC Medicine and I agree with this change.

I have made the following formatting changes as requested:

The title, title page, and affiliations have been changed as requested and the word counts have been removed.
E-mail addresses for the co-authors have been added.
The abstract sub-headings have been changed and the ‘Introduction’ is now a ‘Background.’
The references noted in the manuscript text are bracketed.
A conclusions section has been inserted.
The authors’ contributions have been delineated as requested.
We have no competing interests and have stated so in the manuscript.
The references section includes all the author’s names.
The tables have been inserted at the end of the manuscript along with the Figure title and legend.

Thank you for your attention to this manuscript. Please contact me if additional changes are needed.

Mark E. Anderson MD FAAP
Assistant Professor of Pediatrics
School of Medicine, University of Colorado
Community Health Services, Division of Pediatrics
Denver Health
777 Bannock Street, Mail Code 1911
Denver, Colorado 80204
303 436 4098
manderso@dhha.org